Author's response to reviews

Title: A young man with myelosuppression by clindamycin: A case report.

Authors:

Manuel Morales (drmmoralesp@gmail.com)
Anna Paola Carvallo (eemm37@hotmail.com)
Karla Espinosa (karlaadrianae@gmail.com)
Edgar Murillo (edhob@yahoo.com)

Version: 3 Date: 26 September 2013

Author's response to reviews:

A young man with myelosuppression by clindamycin: A case report.

Morales Polanco Manuel1, Carvallo Thome Anna Paola2, Espinosa Bautista Karla Adriana3, Murillo Meza Edgar Enrique4*.

Address: The American British Cowdray Medical Center. Sur 136, número 116, colonia Las Américas, Delegación Álvaro Obregón, C.P. 01120, México, D.F.
1,2,3,4.

Department of Hematology and Bone Marrow Transplantation 1,3,4.
Department of Internal Medicine 2

Email: drmmoralesp@gmail.com 1
eemm37@hotmail.com 2
karlaadrianae@gmail.com 3
edhob@yahoo.com 4*

Changes made about concerns by the reviewers:

1) We fully re-edited the whole manuscript and put special attention to English grammar.
2) We include the possibility of the infectious state as one associated factor that could explain the pancytopenia.
3) We mentioned the follow-up of the patient 3 months after discharge.
4) We made all the restructure suggested by editor email.